Author's response to reviews

Title: Postoperative outcome of caesarean sections and other major emergency obstetric surgery by clinical officers and medical officers in Malawi

Authors:

Garvey Chilopora (garveychip@yahoo.co.uk)
Caetano Pereira (pecaetano@yahoo.com.br)
Francis Kamwendo (fwkamwendo@hotmail.com)
Agnes Chimbiri (achimbiri@yahoo.co.uk)
Eddie Malunga (emalunga@medcol.mw)
Staffan Bergstrom (Staffan.Bergstrom@ki.se)

Version: 4 Date: 13 June 2007

Author's response to reviews:

According to you e-mail June 12, 2007, we have made the following revisions with changes in the text that we submit now:
1) Author list: corrected
2) Affiliations: all are there for each author.
3) Tables: the text has been changed so that all tables now appear in order.
4) References: no "et al" is there.